Author's response to reviews

Title: PAHA Study: Psychological Active and Healthy Ageing. Psychological well being, proactive attitude and happiness effects of whole-body vibration vs. multi-component training in aged women: study protocol for a randomized controlled trial.

Authors:

Angelo Compare Prof (angelo.compare@unibg.it)
Cristina Zarbo Dr (cristinazarbo@gmail.com)
Elena Marín Dr (elena_mc_90@hotmail.com)
Alessia Meloni Dr (amalessiameloni@gmail.com)
Jacobo A Rubio-Arias Dr (jararias@ucam.edu)
Rosendo Berengüí Dr (rberengui@ucam.edu)
Enzo Grossi Prof (Enzo.Grossi@bracco.com)
Edo Shonin Dr (e.shonin@awaketowisdom.co.uk)
Pedro E. Alcaraz Prof (palcaraz@ucam.edu)

Version: 4
Date: 30 March 2014

Author's response to reviews: see over
Title: PAHA Study: Psychological Active and Healthy Ageing. Psychological well being, proactive attitude and happiness effects of whole-body vibration vs. multi-component training in aged women: study protocol for a randomized controlled trial

POINT-BY-POINT RESPONSE TO REVIEWER’S COMMENTS

We wish to thank the editor and the reviewer for their appreciation of the study and the fruitful comments. A response to their comments is provided below and has been included in the new version of manuscript. Changes in the manuscript have been marked by track revision on the text.

Reviewer request:

The handling editor has completed her assessment of your revisions, and has requested that you revise the introductory section of your paper with the aid of a native English speaker.

Response:

Introductory section has been revised by a native English speaker.